Author’s response to reviews

Title: Relationship between drug application and mortality rate in Chinese older coronary artery disease/chronic heart failure patients with and without low glomerular filtration rate

Authors:

Shihui Fu (xiaoxiao0915@126.com)
Ping Ping (469719796@qq.com)
Ping Ye (sci301@126.com)
Leiming Luo (lleim@sina.com)

Version: 3 Date: 29 Jun 2019

Author’s response to reviews:

BMC Pharmacology and Toxicology

June 21, 2019

Response letter

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript. We appreciate the reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled "Relationship between drug application and mortality rate in Chinese older coronary artery disease/chronic heart failure patients with and without low glomerular filtration rate".

Those comments and suggestions are all valuable and very helpful for us to revise and improve our paper, as well as very important to direct our researches. We study the reviewer's comments carefully and try our best to revise the paper. Attached please find the revised version, which we would like to submit for your kind consideration.

Reviewer 1:

1. Thank you for including a consent to participate section. Please be sure to clearly indicate who you obtained consent from in this section. *** Please also include an ethics and consent statement in the Methods.

We are very sorry for our unclear description. Thank you for your valuable suggestions very much. Your valuable suggestions are very valuable and helpful for us to improve our paper. Written consent was obtained from patients when admission and Helsinki declaration was
followed by the study. Under your guidance, we made the corresponding modification in our manuscript.

2. In the Methods section, please clarify whether this study was a retrospective review of medical records. If not, please clearly indicate the type of study you had conducted in this section.

We are very sorry for our unclear description. Thank you for your kind suggestions very much. Your valuable suggestions are very valuable and helpful for us to improve our paper. As a retrospective review of medical records, the goal of this analysis was to investigate the relationship between drug application and mortality rate in Chinese older CAD/CHF patients with and without low glomerular filtration rate (GFR). Under your guidance, we made the corresponding modification in our manuscript.

3. Please proofread and ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes, colored text, or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

We are very sorry for our unclear description. Thank you for your useful suggestions very much. Your valuable suggestions are very valuable and helpful for us to improve our paper. Under your guidance, we made the corresponding modification in our manuscript.

We have to apologize for giving you so many troubles. We deeply appreciate your consideration and suggestions of our manuscript and look forward to receiving your comments. Your comments and suggestions give us not only great help in revising the article, but also significant revelation in our scientific research. Your kind guidance is our good luck. We wonder if the modification could meet your requirements. If you have any queries, please don’t hesitate to contact us.

Thank you and best regards.